Author’s response to reviews

Title: Safety and Tolerability of Experimental Hookworm Infection in Humans with Metabolic Disease: Study Protocol for a Phase 1b Randomised Controlled Clinical Trial

Authors:
Doris Renate Pierce (doris.pierce@my.jcu.edu.au)
Lea Merone (lea.merone@my.jcu.edu.au)
Chris Lewis (chris.lewis2@sa.gov.au)
Tony Rahman (tony.rahman@health.qld.gov.au)
John Croese (jcroese@bigpond.com)
Alex Loukas (alex.loukas@jcu.edu.au)
Malcolm McDonald (malcolm.mcdonald@jcu.edu.au)
Paul Giacomin (paul.giacomin@jcu.edu.au)
Robyn McDermott (robyn.mcdermott@jcu.edu.au)

Version: 4 Date: 21 Nov 2019

Author’s response to reviews:

BEND-D-19-00021R3: Response to Editor

1. Trial Registration Number
Thank you for including a link to your Trial on the ANZCTR Website. We note the Trial Registration Number is incorrect, on the link you provided number is listed as: ACTRN12617000818336 and in your manuscript as: ACTRN1267000818336. Please amend this so that it is searchable and matches with the ANZCTR website.

Response: Apologies for the incorrect number; this has been amended now.

The last section of the abstract should be Trial Registration: listing the trial registry and the unique identifying number, e.g. Trial registration: Current Controlled Trials ISRCTN73824458, as well as the date of registration. Please note that there should be no space between the letters and numbers of the trial registration number.
2. Clean manuscript
Please remove any files from your file inventory that you do not wish to be published alongside your manuscript.

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: Manuscript and all other files are clean